IZA DP No. 121

Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?

Jan C. van Ours

March 2000

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?

Jan C. van Ours

Tilburg University, The Netherlands, CentER for Economic Research, Institute for Labour Studies (OSA), The William Davidson Institute, CEPR and IZA, Bonn, Germany

Discussion Paper No. 121 March 2000

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research areas *Labor Markets in Transition* and *Project Evaluation*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets and European integration, (3) the welfare state and labor markets, (4) labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 121 March 2000

ABSTRACT

Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?^{*}

This paper uses an administrative dataset to analyze to what extent active labor market policies in the Slovak Republic have been beneficial for unemployed workers. The focus is on two types of temporary subsidized jobs and on training. Short-term subsidized jobs seem to be the most efficient active labor market policy. Workers that are or have been on a short-term subsidized job have a higher job finding rate than other unemployed workers have and once they find a job they have a lower job separation rate than workers that have not been on a short-term subsidized job. Long-term subsidized jobs have a negative effect on the job finding rate and no effect on the job separation rate. The positive effect of training on the job finding rate of unemployed workers may have to do with reversed causality: some workers enter a training program only after they are promised a job. Training does not seem to affect the job separation rate.

JEL Classification: J64, C41

Keywords: Unemployment, active labor market policy, duration models

Jan C. van Ours Department of Economics Tilburg University P.O. Box 90153 NL-5000 Tilburg The Netherlands Tel.: +31-13-4662316 Fax: +31-13-4663042 Email: j.c.vanours@kub.nl

^{*} The author is grateful for the financial support from the Volkswagen Foundation and Phare Ace Program of the European Commission and for the help of Martina Lubyova in collecting the data and in providing information about the institutional characteristics of the Slovak labor market. The author thanks participants of the June 1999 ZEW-conference on Econometric Evaluation and the November 1999 IZA-CEPR workshop on program evaluation, in particular Denis Fougère for comments on a previous version of the paper.

1 Introduction

Active labor market policies (ALMP) aim to improve the labor market position of unemployed workers. In many countries training and subsidized jobs are important elements of ALMP programs. Subsidized jobs, both in the market sector and the government sector affect the job opportunities of unemployed workers because of the increased number of jobs available. Furthermore, they may upgrade individual unemployed workers when there is a training element involved. The knowledge from training courses and the work experience that unemployed workers collect on a subsidized job may also make them more valuable for employers. Employers in search for new employees may prefer workers who have been on a training course or worked on temporary subsidized jobs to other unemployed workers. If they find a regular job more quickly training and temporary subsidized jobs can be considered as a "treatment" for unemployed workers.

This paper studies the treatment effects of training programs and subsidized jobs that were part of the Slovak program of ALMP. The focus is on the individual treatment effects. So there is no evaluation of possible crowding out phenomena by which treatment may increase the job finding rate of "treated" unemployed workers but only at the expense of non-treated unemployed workers. Nor is there an evaluation of the dead weight loss of the temporary subsidized jobs, which is that some of the treated unemployed workers would have found a job even without treatment.

Slovakia has about 2.2 million employed workers and a 1997 GDP per capita (purchasing power parity) of \$8,600. The Slovak labor market is one of the transitional labor markets of Central and Eastern Europe (see Svejnar (1999) for an overview). Like many other countries with a transitional economy, Slovakia experienced a sharp increase of unemployment at the initial stage of the transition. In the course of 1991, unemployment increased from practically zero to 300,000 persons which corresponded to an unemployment rate of about 12% (OECD 1996).

In the course of the 1990s the Slovak government introduced a range of ALMP. Important elements of these ALMP were temporary subsidized jobs known as socially purposeful jobs (SPJ) and publicly useful jobs (PUJ). SPJ were mainly created in the private sector and concerned higher qualified functions while PUJ were low ranking jobs in the public sector best described as "community works" (OECD (1996)). ALMP were introduced in 1991 and gradually developed into a comprehensive system of more than eight basic types of programs. Until the end of 1996 the ALMP in Slovakia contained the following measures: SPJ, PUJ,

retraining, counseling, sheltered workplaces for the disabled, subsidies for shortened working hours and places for school-leavers. SPJ were the most important throughout the period, both in terms of number of created jobs and expenditures. They were followed by PUJ and retraining. The last major reorganization occurred in 1997 when SPJ and PUJ were combined into one program of subsidized jobs.

In Lubyova and Van Ours (1999) the effects of PUJ, SPJ and training on the transition rate from unemployment to a regular job are investigated. The results of this study are discussed in more detail below. The main purpose of the current paper is to study the effects of ALMP more closely, both by extending the previous analysis and by investigating additional effects. I investigate whether the exit rate to regular jobs increases if an unemployed enters a PUJ, a SPJ or a training program. Furthermore, I investigate whether the separation rate from a new job is related to whether or not the worker previously participated in an ALMP. I use a sample of 3 Slovak districts of which I have detailed labor market information with respect to the workers that started their unemployment spell in 1993. I focus on the labor market position of male workers.

The paper is set up as follows. Section 2 provides stylized facts about the system of passive and active labor market policies in Slovakia. I also discuss the results from a previous study on the effects of Slovak ALMP. Section 3 describes the data and the sampling procedure. Section 4 presents the statistical model and Section 5 presents the estimation results. Section 6 concludes.

2 Labor market policies

The Slovak Republic has a well-educated labor force. In 1994 only 14% of the Slovak labor force had primary or less education, 74% had secondary education and 12% had a high education (OECD (1996)). Furthermore, participation rates are high. The participation rate of males in 1994 was 83% and of females 71%.

There are big differences with respect to the extent to which Slovak workers are confronted with unemployment. In 1994 the unemployment rate of workers aged 15-24 was 29%, for workers aged 25-49 it was 12%, while for the age group 50-59 it was 8%. Furthermore, there are big differences between educational categories with unemployment rates being lower for higher educated workers: The unemployment rate of workers with basic or no education was 28%, while for academics it was 3%. Finally, there are big differences between regions, with the unemployment rate of 4% in Bratislava being the lowest and rates of 25-30% in the Eastern parts of Slovakia being the highest.

The system of unemployment benefits in the Slovak Republic has been discussed elsewhere (for a detailed description of institutions I refer the reader to OECD (1996)). Therefore, I only give a summary of the key features of the system.

In 1993 workers who lost their job were entitled to unemployment benefits if they were registered at the labor office and had 12 months of work experience during the previous 3 years. The entitlement period lasted 6 months. In 1995 the entitlement period was made dependent on the age of the worker.

Unemployed who exhausted their unemployment benefits eligibility or who were not eligible for unemployment benefits or whose household income did not reach the minimum guaranteed income while collecting unemployment benefits were entitled to social assistance benefits. Eligibility was based on a means test and a benefit was only given when the household income was below the minimum living standard (MLS). The structure of social assistance benefits was the same for unemployed and others. For members of households consisting of two adults the replacement ratio in case of a minimum wage job was about 80%. As soon as children are involved the replacement ratio for minimum wage jobs was 100%. Even for average wage jobs the replacement ratio for adults with children and unemployment benefits was no less than about 80%. After transfer to the social assistance benefits their replacement ratio was about 50-60% (for more details see Lubyova and Van Ours (1997, 1998)).

The remainder of this section gives a brief description of active labor market policies and of the evaluation studies on the effects of ALMP.

2.1 Active labor market policies

Active labor market policies consist of a range of measures from subsidized jobs to training. We focus here on SPJ, PUJ and training. The SPJ program has been the most extensive one among Slovak ALMP programs both in terms of expenditures and participants. SPJ were subsidized jobs in the private sector. The concept of SPJ and the rules of administration have undergone numerous revisions as the authorities learned how to tailor the programs to labor market conditions. In 1991 SPJ were considered to be every job created on the basis of an agreement with the labor office by an employer in production, business or other activities aimed at making profits. In 1992 the profit-seeking requirement was eliminated and the requirement that the job had to be occupied by registered unemployed was introduced. The latter was partially relaxed in 1994 when the school-leavers, persons younger than 18 years and those who would be full-time self-employed under SPJ were allowed to participate without prior registration. The main forms of support introduced in 1991 were subsidies, interest repayments and loans, later reduced to 2-years loans and subsidies. The minimum duration of SPJ was introduced in 1992 and set to 2-years period. In case of lay-off or quit, the job had to be occupied by another registered unemployed within 30 days.

Publicly useful jobs were designed mostly for lower qualified workers for a limited period of time. In 1991 PUJ were introduced as short-term employment opportunities created on the basis of agreements between labor offices and non-

profit employers (for example, organs of state administration, municipalities, and local administration). The requirement for non-profit orientation of the employer was canceled in 1992. State budgetary organizations and state contributory (partial budgetary) organizations were excluded from PUJ programs in 1994. The upper limit for financial support was originally set at the wage costs of the participant, later extended to cover also participant's social insurance contributions. The maximum duration of PUJ in 1991 was 6 months. Given that the participation renewed unemployment benefit entitlement, many unemployed workers were shifting between PUJ and open unemployment. Therefore, the maximum duration of PUJ was raised to 9 months in 1994 and to 12 months in 1995. The stocks were strongly built up after two major inflows of about the same size, which occurred in the financing boom of 1992, and in the first half of 1995. The latter inflow was a result of changed priorities in 1995 - more means were put into PUJ, partly at the expense of other programs.

Training of unemployed workers has been used relatively modestly. Where possible, retraining of employees in enterprises was preferred to retraining of unemployed by labor offices. The average share of unemployed in the total number of retrained persons was typically 60 to 70%; the share of retrained persons in the total number of unemployed was typically less than 5%. In 1994 the average duration of a retraining course was 8 weeks, the overall range spanning 4 to 19 weeks. The rate of subsequent job-placement was rather high. This may be due to the ex-ante promise of job-placement required by many labor offices in the light of severe shortage of available vacancies.

The implementation of ALMP was in the hands of the Public Employment Service (PES) that had a network of district offices where every district office had a number of local centers. So, the services were never far away (OECD (1996)). Although priority of placement was given to long-term unemployed workers, the target group of the wage subsidies was not limited to the long-term unemployed. Every unemployed person who could not get a normal job was entitled to a subsidized job offered through the PES system. According to the OECD (1996) the incentive to establish a subsidized job usually came from interested employers. Among the subsidized jobs were jobs that required no special training and education, including caretaking, cleaning, kitchen work and unskilled jobs in general. The creation of subsidized jobs was a matter of negotiation between employers and PES. The wage subsidy was granted to individuals but paid to the employer. The wage was comparable to other workers that had a low skilled industrial job and was usually at the minimum wage or somewhat above. This means that for some workers the replacement rate was quite high. If someone refused a job offered by the PES he or she may have gotten a benefit sanction imposed but the labor offices were usually reluctant to use this instrument.

In 1997 the structure of ALMP programs was substantially reformed. A new act on employment, effective since January 1997, stipulated the following ALMP measures: retraining, support of job creation, support of employment of specific

target groups, support for prevention of layoffs, support for maintenance of existing jobs, support for elaborating proposals for revitalization of employment, support for employment of disabled. More attention was given to the measures of a preventive nature, as well as to targeting of the measures to specific groups (disabled, older people, long-term unemployed, young workers and school leavers). The original SPJ and PUJ were formally unified into one program of subsidized jobs, although some distinction between the two types of jobs was preserved.

2.2 Evaluation of the effects of ALMP

There have been some studies on the impact of ALMP in transition economies, but there is not an abundant number. From an overview of studies on labor-market reforms in transition economics Boeri (1997) concludes that active policies, such as subsidized employment schemes and public work programs have not been very successful. According to Boeri this may have to do with the phenomenon that slots in training courses are often offered to job seekers with rather favorable labor market characteristics who would have found a job anyway. Furthermore, participation in ALMP may stigmatize the participants, which will reduce their chances of finding a regular job.

With respect to the subsidized jobs in Slovakia the long-term effects for participating workers are considered to be unclear. According to the OECD (1996) it is not clear to what extent workers that have a subsidized job requalify for later employment is the competitive sector. Nevertheless the OECD (1996) expresses some hope of positive effects from the simple fact that subsidized workers hold a normal position in society as long as the subsidy lasts.

Lubyova and Van Ours (1999) investigate to what extent it is beneficial for unemployed workers who want a regular job to enter an ALMP-program, which is defined as an SPJ, a PUJ or a training program. They do separate analyses for males and females. Their estimation results indicate that female, lower educated and older unemployed workers have a worse labor market position, both in terms of the speed with which they find regular jobs as in terms of the speed with which they enter ALMP-programs. Those workers who have a better position when it comes to finding regular jobs are also in a better position to enter an ALMPprogram. The ALMP-programs seem to be complementary to the regular labor market rather than compensating for bad labor market characteristics. Furthermore, it turns out that selectivity is important in the transition processes. For both males and females there are two groups of apparently identical individuals that have different transition rates to both regular jobs and ALMPprograms. There is one group of workers of about 10% that has a low transition rate to a regular job and a high transition rate to an ALMP-program. Another group of unemployed workers of about 90% has a higher transition rate to a regular job and a low transition rate to an ALMP-program. Selectivity in the

transition to ALMP-programs is an important phenomenon. If selectivity is not accounted for the estimated treatment effect of ALMP-programs is negative. If selectivity in the inflow into ALMP-programs is accounted for, it is found that workers that enter these programs benefit from it. After entering an ALMP-program the exit rate to a regular job increases with 150%. From an additional sensitivity analysis Lubyova and Van Ours (1999) conclude that this effect is mainly due to the positive effect of training and PUJ. For SPJ they find a negative effect on the transition rate from unemployment to a job. The analysis in Lubyova and Van Ours (1999) is limited in the sense that there is no distinction between the transition rates to the separate ALMP-programs. The large positive effect of training may come from the fact that unemployed that enter a training program get the ex ante promise of a regular job. Then, training is not a treatment to increase the transition rate from unemployment to a job, but the promise of a job is a 'treatment' to increase the inflow into training.

3 Data

The data used in our analysis come from the unemployment registers of labor offices in three Slovak districts. The first district is Bratislava (excluding the capital Bratislava) with a December 1993 unemployment rate of 4.1%, the second district is Dolny Kubin with an unemployment rate of 12.9% and the third district is Bardejov with an unemployment rate of 19.3%.

In the selected districts the data collection was exhaustive, i.e. all the registered unemployed were selected. Several types of information are used in order to reconstruct individual histories. An individual history consists of a sequence of spells representing three possible labor market states: employment, unemployment and out of labor force. In addition to that the spells of participation in SPJ and PUJ programs are identified. The information about the last unemployment spell comes directly from the unemployment register, which contains also information about the personal characteristics of job applicants. Due to the overwriting of records in case of repeated registration, the unemployment register does not contain information on previous unemployment history. Therefore, unemployment archives had to be used in order to reconstruct the information, separate files are used that contain information about previous employment or out of the labor force spells prior to every registration. Finally, information about dates and duration of ALMP programs are used¹.

From the unemployment register and unemployment archives an inflow sample was selected of all the unemployed that became registered in the course of 1993. The censoring point is April 1998. By combining the information from the

¹ The analysis is restricted to SPJ, PUJ and training, ignoring other ALMP programs.

unemployment register the unemployment archives and information on ALMP, individual histories were constructed, each ending at the above-mentioned censoring date. The use of 1993 inflow is justified by the relative stability in the institutional set-up of the labor market (major reforms occurred at the beginning of 1992 and 1995). Another reason was a sufficient time period elapsed before the censoring point in order to avoid large shares of censored spells. In the analysis I use information about the length of the first spell of unemployment that started in 1993 and if this spell ended about the labor market status after unemployment. If the spell ended in a transition to a job the unemployment spell was considered to be completed. If the spell ended in a transition to an ALMP-program the unemployment duration was considered to continue until another transition occurred either to a job or back to unemployment. When a transition to a job happened the unemployment spell was considered to be complete. When a transition occurred back to unemployment the spell was still considered to be incomplete. In the analysis the duration of unemployment up to a transition to an ALMP-job or to training is also important. This duration is the search period until an ALMP measure is met. If the spell did not end or ended in a transition to out of the labor force the unemployment spell is considered to be right censored.

State 1	State 2	State 3	State 4
U (5057)	PUJ (332)	E (128)	U (91)
			C (37): E (26), c (11)
		C (204): unknown (7), SPJ (1),	
		PUJ (40), U (114), c (42)	
	SPJ (275)	E (177)	U (87)
			C (90): SPJ (1), E (52),
			c (37)
		C (98): unknown (30), SPJ (3),	
		U (61), c (4)	
	Tr (179)	E (132)	U (70)
			C (52): SPJ (5), E (40),
			c (17)
		C (47): unknown (20), SPJ (11),	
		PUJ (5), U (5), Tr (6)	
	E (3525)	U (2452)	
		C (1073): unknown (1),	
		E (776), c (296)	
	C (746): unknown (462)		
	U (240), c (44)		

Table 1 Transitions in the sample; numbers of workers in different subsequent labor market $states^{a)}$

^{a)} E = regular job, U = unemployed, PUJ = publicly useful job, SPJ = socially purposeful job, Tr = training, C = considered to be censored, c = actually censored

The explanatory variables I use in the analysis refer to age, education and marital status. Appendix 1 gives details about our dataset. As shown there, on average 35% of the unemployed in our sample was younger than 30 years, 59% had a lower education and 58% was married. Apart from the unemployment rate the largest differences between the unemployed in the three districts seem to be age or age-related. In Bratislava on average the unemployed are older than in the other two districts, they are lower educated and have a higher share of married man.

Table 1 gives some indication about the transitions in labor market status that were used in the analysis. Over the period of observation 332 unemployed entered a PUJ, 275 entered a SPJ, 179 entered a training program and 3525 unemployed found a regular job without having been on an ALMP program. For 746 unemployed the unemployment spell is right censored. There are several reasons for the spells to be considered as right censored. Some of them really are right censored in the sense that at the end of the sample period they were still unemployed. Other workers have left unemployment with unknown destination. Then, there are people who left unemployment and re-entered immediately, probably for administrative reasons.

Furthermore, Table 1 shows that of the 332 unemployed workers that have been on a PUJ 128 entered a regular job, 177 entered a regular job after being on a SPJ and 132 entered a regular job after being on a training program. The remaining spells are considered to be right censored unemployment spells. A lot of unemployed workers that have found a regular job during the sampling period have lost their job and reentered unemployment. There are 776 workers that have found a new job after their first one. The estimations are based on the duration of the first job which we assume to be right censored at the start of the second job. Note that in total only 9% of the transitions go to an unknown labor market state.



Figure 1 Quarterly transition rates; three Slovak districts

Figure 1 shows quarterly transition rates. From this figure it appears that apart from a peak in the second quarter the transition rate from unemployment to a regular job is sort of constant over the duration of unemployment. The entrance rate into ALMP programs increases over the first eight quarters and fluctuates after that. The transition rate from a regular job back to unemployment has a peak in the third quarter but is sort of constant before and after the third quarter. The transition rates from unemployment to a regular job and from a regular job back to unemployment have about the same magnitude.

4. Statistical model

We consider an ALMP program as a 'treatment' that may affect the behavior of the unemployed workers. In order to establish the effect of an ALMP program on the exit rate from unemployment to a regular job we have to set-up a model that accounts for possible selectivity in the inflow into ALMP. Heckman et al. (1999) gives an overview of the relevant issues when one wants to establish treatment effects. There have been many treatment effect studies but studies that estimate the effects of the treatment in the context of an event history model of labor force dynamics are rare². In our model we exploit information with respect to the duration of unemployment, the duration of the stay in an ALMP, the destinations after that and the duration of a subsequent employment spells. In multivariate duration models the variation in the durations at which treatment is administered to individuals, and data on the corresponding pre- and post-treatment durations can be exploited to identify the treatment effect. The intuition is as follows. Consider the transition rate from unemployment to a regular job, which is affected by unobserved heterogeneity that has a discrete distribution with two points of support. Consider also the transition rate from unemployment to an ALMP that is affected by a similar type of unobserved heterogeneity. If the two types of unobserved heterogeneity are correlated this means that conditional on observed characteristics there are four groups of individuals that differ in terms of transition to a job (high/low) and transition to an ALMP (high/low). Conditional on observed characteristics each of these four groups is homogenous. So, within the groups selectivity of the inflow into an ALMP cannot be present. Therefore, we are able to estimate the unbiased effect of the ALMP on the transition rate from unemployment to a regular job if we can identify the unobserved heterogeneity in both the transition rate to a regular job and the transition rates to ALMP. This means that the data should contain information about all relevant transitions over some period of time, which they do^3 .

 $^{^{2}}$ An early example of the use of duration models in an evaluation study is Ridder (1986).

³ If all unemployed would enter an ALMP at the same elapsed duration of unemployment we could not identify potential selectivity.

A formal proof of the identification of the treatment effect is given in Abbring and Van den Berg (1998). Van den Berg (2000) presents an overview of duration models and has a general discussion on the use of duration models in estimating treatment effects. He indicates that in order to avoid biased estimates of the impact of ALMPs on unemployment durations it is necessary that individuals do not behave in anticipation to future events.

Examples of the use of multivariate duration models in evaluation studies are Abbring, Van den Berg and Van Ours (1997) and Van den Berg, Van der Klaauw and Van Ours (1998)⁴. In these studies the effect of benefit sanctions on the transition rate from unemployment to employment is modeled. Here too, the issue of selectivity is very important. Again, selectivity is accounted for by simultaneously modeling the exit rate out of unemployment and the rate by which unemployed get a sanction imposed. Both studies find a significant positive effect of benefit sanctions on the transition rate from unemployment to a job. In the study by Van den Berg, Van der Klaauw and Van Ours (1998) it is shown that if unobserved heterogeneity is not accounted for, no effect of sanctions is found. Furthermore, if only the transition rate from unemployment to a job is taken into consideration the presence of unobserved heterogeneity cannot be detected. Therefore, it seems to be important to take the possible correlation between the unobserved components of the transition rate to a job and the transition rate to the sanction state into account.

I do not only consider the effect of ALMP on the transition rate from unemployment to a job but also consider the effect of ALMP on the separation rate once workers have found a job. In this respect the current paper is in the line of Gritz (1993) and Bonnal et al. (1997). Gritz (1993) considers the impact of training on the employment experience of American youths. He distinguishes between three labor market states, unemployment, employment and training and estimates the coefficients of the six related transition rates. To account for selectivity he models unobserved heterogeneity assuming a discrete distribution with a finite number of points of support. He finds five points of support to be optimal, so there appear to be five classes of individuals in the unobserved heterogeneity distribution. Bonnal et al. (1997) study the effect of public employment policies set up in France during the 1980's. These policies intended to improve the labor market position of young unskilled workers. Bonnal et al. use individual labor market transition data distinguishing between six labor market states: unemployment, permanent job, temporary job, public employment policy job, out-of labor force, and attrition. This means that there are 25 possible labor market transitions. Their statistical model is of the mixed proportional hazard (MPH) type, with piecewise constant duration dependence. To reduce the complexity of the model restrictions are imposed on the distribution of unobserved heterogeneity. Two alternative specifications of the distribution of unobserved

⁴ A related study is Holm, Van den Berg and Van Ours (1998).

heterogeneity are investigated. In the first specification the unobserved heterogeneity components are generated by a common normally distributed random variable. In this case the correlation between the unobserved components can only equal 0, 1 or -1. In the second specification the unobserved components are assumed to have discrete multivariate distributions with a finite number of points of support. Because a six-dimensional discrete distribution involves a large number of parameters Bonnal et al. (1997) use a two-factor loading model with four points of support. The second specification is much more flexible than the first. From their estimation results it appears that participation in labor market programs is highly selective in the model with the second specification of unobserved heterogeneity, while no selection effect is found in the model with the first specification of unobserved heterogeneity. Apparently, a flexible specification of unobserved heterogeneity is very important.

To establish the effect of selectivity I start with a simple duration model and subsequently introduce more complexity. The baseline model has for every transition rate a proportional specification with a flexible baseline hazard. Differences between unemployed individuals in the transition rate from unemployment to a job can be characterized by the observed characteristics x, the elapsed duration of unemployment itself⁵, and a variable indicating whether or not the individual started participating in an ALMP. I assume x to be time-invariant. Furthermore, t_a (a = PUJ, SPJ, retraining) is the time at which the individual starts participating in an ALMP and I(t_a<t) is the dummy variable indicating whether the individual has already started participating. I use a similar specification for the transition rate to PUJ, SPJ and training and for the transition rate from regular jobs back to unemployment. In the latter transition rate I include a dummy variable d_a that indicates if a worker has participated in an ALMP program $(d_a = 1)$ or not $(d_a = 1)$ = 0). The transition rate from unemployment to a regular job at time t conditional on x and t_a, the transition rate to SPJ, PUJ or training at time t conditional on x and the transition rate from a regular job back to unemployment at time t conditional on x and d_a can be specified as follows:

$$\begin{split} \theta_{ue}(t; x, t_a) &= \lambda_{ue} (t).exp(x'\beta_{ue} + \delta_{ue,a}.I(t_a < t)) \\ \theta_a(t; x) &= \lambda_a (t).exp(x'\beta_a) \end{split} \tag{1}$$
$$\theta_{eu}(t; x, d_a) &= \lambda_{eu} (t).exp(x'\beta_{eu} + \delta_{eu,a}.d_a) \end{split}$$

where the $\lambda_j(t)$ -functions, j = ue, a, eu represent individual duration dependence, $\delta_{ue,a}$ measures the effect that participation in an ALMP has on the transition rate from unemployment to a regular job and $\delta_{eu,a}$ measures the effect of previous

⁵ As indicated above in the calculation of the unemployment duration I add the time spend during ALMP. So, if t_1 is the unemployment spell until entering an ALMP-program, t_2 is the time spend in the program and t_3 is the time unemployed after the program and before entering a regular job, total unemployment duration $t = t_1+t_2+t_3$.

participation in an ALMP on the separation rate from a regular job. In both cases I assume the treatment to be an incidence effect (Gritz (1993))⁶. There could be a lot of aspects of the ALMP that potentially affect the transition rate to a regular job and the job separation rate, but I only take the effect of the participation in an ALMP into account. Flexible duration dependence is modeled by using step functions:

$$\lambda_{j}(t) = \exp(\Sigma_{k}(\lambda_{j,k} I_{k}(t))$$
(2)

where j = ue, a, eu is an indicator for the type of transition, k (= 1,..,4) is a subscript for time-intervals and $I_k(t)$ are time-varying dummy variables that are one in subsequent time-intervals. I distinguish four time intervals: 1-2 quarters, 3-4 quarters, 4-8 quarters and 8+ quarters⁷. Because a constant term is also estimated I normalize $\lambda_{ue,1} = \lambda_{a,1} = \lambda_{eu,1} = 0$.

The conditional density functions of the completed unemployment durations t_u , the completed durations until entrance of an ALMP t_a and the completed employment durations t_e can be written as

$$f_{ue}(t_{u}; x, t_{a}) = \theta_{ue}(t_{u}; x, t_{a}) \exp(-_{0} \int^{tu} \theta_{ue}(s; x, t_{a}) ds)$$

$$f_{a}(t_{a}; x) = \theta_{a}(t_{a}; x) \exp(-_{0} \int^{ta} \theta_{a}(s; x) ds)$$

$$f_{eu}(t_{e}; x, d_{a}) = \theta_{eu}(t_{e}; x, d_{a}) \exp(-_{0} \int^{te} \theta_{e}(s; x, d_{a}) ds)$$
(3)

Since we have an inflow sample and the transition rates are uncorrelated, the loglikelihood L of the model factorizes where the components are specified as

$$L_{j} = d_{j} \sum \log(f_{j}) + (1 - d_{j}) \sum \log(1 - F_{j})$$
(4)

where the F_j 's are distribution functions, d_j is a dummy variable with a value of 1 if the transition j is completed and a value zero is the duration of the spell is right censored.

In the second model I allow for unobserved heterogeneity to affect the transitions to both a regular job, to one of the ALMP programs and from a job back to unemployment:

$$\begin{aligned} \theta_{ue}(t; x, u, t_a) &= \lambda_{ue} (t).exp(x'\beta_{ue} + \delta_{ue,a}.I(t_a < t) + u) \\ \theta_a(t; x, v_a) &= \lambda_a (t).exp(x'\beta_a + v_a) \\ \theta_{eu}(t; x, w, d_a) &= \lambda_{eu} (t).exp(x'\beta_{eu} + \delta_{eu,a}.d_a + w) \end{aligned}$$
(5)

⁶ Note that because of this the changes in the length of PUJ-jobs that occured in the course of time do not have to be taken into account explicitly.

⁷ The intervals were chosen because they are related to the social security system (the workers that became unemployed during 1993 had a UB-entitlement period of 6 months) or because of convention (the border line between short-term and long-term unemployment is 1 year, the border line between long-term and very-long-term is 2 years).

where u, v_a and w are the components of unobserved heterogeneity in the transitions. These components are assumed to follow a discrete distribution with two points of support (p_j and $1-p_j$). In the second model I assume the unobserved heterogeneity components to be independent. Therefore the conditional density functions and the loglikelihoods are similar to those in (3) and (4).

The basic assumption so far is that the inflow into the ALMP programs is a random process in the sense that it is independent of the process by which unemployed find jobs and subsequently loose jobs. The selection into the treatment-program is assumed to be exogenous and not dependent on unobserved characteristics that also affect the job finding rate or the job separation rate. In other words, conditional on observed characteristics and the duration of unemployment the quality of the unemployed flowing into an ALMP is as good (or as bad) as the quality of the unemployed workers that remain unemployed. Then, if there is an effect of an ALMP ($\delta_{ue,a} \neq 0$), this is a 'true' effect. This effect can go both ways. If $\delta_{ue,a} < 0$ the ALMP has a negative effect on the reemployment hazard, which could be caused by stigmatization. If $\delta_{ue,a} > 0$ the workers on an ALMP have a higher exit rate to a job than the non-participants. Note that in the specification of the hazard in equations (1) and (5) the effect of an ALMP occurs immediately. Also note that I consider the duration of a stay in an ALMP as extended unemployment duration. This concept does not coincide with the official statistics but I take the point of view of a labor economist: a person is unemployed until he or she finds a regular job or leaves the labor market. A similar line of reasoning holds with respect to possible correlation between the flow into an ALMP and the separation rate from regular jobs. In the first two models I assume that there is no such correlation. Again, if $\delta_{eu,a} < 0$ the ALMP has a negative effect on the separation hazard and thus a positive effect on job tenure. Signaling could cause this effect. Workers that have accepted an ALMP signal their willingness to work. If $\delta_{eu,a} > 0$ the workers on an ALMP have a higher job separation rate than the non-participants, which could be caused by the fact that the match between workers from an ALMP and a regular job is less efficient.

In the third model the unobserved heterogeneity terms are allowed to be correlated. Now selectivity in the inflow into one of the ALMP programs is accounted for. If unobserved characteristics have a negative effect on the job finding rate and a positive effect on the transition rate to an ALMP, then conditional on the observed characteristics and the elapsed duration of unemployment the average quality of the workers in an ALMP is lower than the average quality of workers who do not enter an ALMP. Then, if one would simply compare the transition rates to regular jobs of both groups one compares workers with unfavorable characteristics and ALMP with workers with more favorable characteristics and no ALMP. Therefore, one would underestimate the true effect of participating in an ALMP. The opposite effect is also possible. One can imagine that the people in control of the entrance to ALMP want their programs to be a success. Therefore they prefer workers with good characteristics to flow into

their program. This implies that there is a positive correlation between the unobserved heterogeneity components in both transition rates. Then, the effect of an ALMP program is overestimated. $G(u,v_a,w)$ is defined to be the joint distribution of the unobserved characteristics u, v_a and w. Then, the joint density function of t_{ue} , t_a and t_{eu} conditional on x equals

 $f_{ue,a,eu}$ (t_{ue}, t_a, t_{eu}| x; u, v_a, w, t_a, d_a) =

$$\int_{u} \int_{va} \int_{w} f_{ue} (t_{ue} | x, u, t_{a}) f_{a} (t_{a} | x, v_{a}) f_{eu} (t_{eu} | x, w, d_{a}) dG(u, v_{a}, w)$$
(6)

Each of the error terms is assumed to follow a discrete distribution with two points of support, and each error term can be correlated to another⁸. Therefore, G is a discrete distribution of unobserved heterogeneity with eight points of support (u^a , v_a^a , w^a), (u^b , v_a^b , w^a), (u^a , v_a^a , w^a), (u^b , v_a^a , w^b), (u^b , v_a^b , w^b), (u^a , v_a^b , w^a), (u^b , v_a^a , w^b), (u^b , v_a^b , w^b), (u^a , v_a^b , w^a), (u^b , v_a^a , w^b). The associated probabilities are denoted as follows:

$Pr(u=u^{a}, v=v_{a}^{a}, w=w^{a}) = p_{1}$	$Pr(u=u^{b}, v=v_{a}^{b}, w=w^{a}) = p_{2}$
$Pr(u=u^{a}, v=v_{a}^{b}, w=w^{a}) = p_{3}$	$Pr(u=u^{b}, v=v_{a}^{a}, w=w^{a}) = p_{4}$
$Pr(u=u^{a}, v=v_{a}^{a}, w=w^{b}) = p_{5}$	$Pr(u=u^{b}, v=v_{a}^{b}, w=w^{b}) = p_{6}$
$Pr(u=u^a, v=v_a^b, w=w^b) = p_7$	$Pr(u=u^{b}, v=v_{a}^{a}, w=w^{b}) = p_{8}$ (7)

where $0 \le p_i \le 1$ and $\sum_i p_i = 1$, i = 1,...,8. The set-up of the likelihood is similar to the one presented in equation (4). Note that because of the introduction of correlated unobserved heterogeneity it is not possible to factorize the likelihood as we did for the previous models, so we have to estimate all the coefficients jointly.

5 Estimation results

5.1 Parameter estimates

Of all the workers in our sample we have information with respect to the length of the unemployment spell, the destination after the unemployment spell, the length of the subsequent spell, et cetera. On the basis of this information we estimate the coefficients of the models presented in the previous section using the method of maximum likelihood. Apart from the personal characteristics age, education and marital status we also include fixed effects for two of the three districts in the analysis.

Table 2 presents the estimation results for the baseline model. The parameter estimates for the transition rate to a job indicate that age is not relevant for this transition. Furthermore, low educated unemployed workers have a smaller

⁸ Note that I assume that there is perfect correlation between the unobserved components of the three ALMP-transition rates. This correlation may be perfectly negative of perfectly positive.

transition rate to a job than workers with an incomplete, secondary or higher education. Married men have a higher job finding rate than their counterparts. In the districts with a high unemployment rate the direct transition to a regular job is smaller than in districts with a lower unemployment rate. There is also some negative duration dependence in the transition rate from unemployment to a job. However it is only the first six months in which the transition rate is significantly higher than later on. Beyond six months the job finding rate is sort of constant. Finally, it appears that unemployed that have entered a PUJ have a significantly higher transition rate to a job than those that did not have a PUJ. Unemployed that entered a SPJ have a lower transition rate to a regular job than those that did not. Unemployed that entered a training program have a substantially higher transition rate to a regular job.

	To job	To PUJ	To SPJ	To training	from job
Age <30	-0.01 (0.2)	-0.06 (0.5)	0.30 (2.1)*	0.43 (2.3)*	-0.13 (2.8)*
Inc. Sec. ed.	0.20 (3.7)*	-0.33 (1.5)	0.74 (4.1)*	0.47 (2.1)*	-0.15 (2.6)*
Sec higher ed.	0.13 (3.4)*	-0.44 (2.7)*	0.59 (4.1)*	0.70 (3.9)*	-0.28 (5.9)*
Married	0.17 (4.6)*	-0.22 (1.8)	0.43 (2.9)*	0.63 (3.4)*	-0.13 (2.8)*
District 2	-0.08 (2.1)*	-0.02 (0.1)	-0.10 (0.7)	-0.11 (0.6)	0.00 (0.1)
District 3	-0.54 (10.9)*	-0.12 (0.7)	-0.32 (1.8)	0.17 (0.8)*	0.15 (2.9)*
Mass points					
u ^a , v ^a , w ^a	-2.46 (53.7)*	-6.80 (25.4)*	-5.46 (29.0)	* -6.76 (25.2)*	-2.75 (48.2)*
Duration depende	ence				
3-4 Quarters	-0.25 (5.8)*	1.75 (6.4)*	-0.42 (2.6)*	0.42 (2.1)*	0.79 (18.5)*
4-8 Quarters	-0.35 (7.8)*	2.69 (10.5)*	-0.56 (3.2)*	0.60 (3.0)*	-0.09 (1.5)
8+ Quarters	-0.40 (7.2)*	3.41 (13.4)*	-0.65 (2.4)*	0.58 (2.2)*	0.09 (1.1)
Treatment effects					
PUJ	0.41 (4.2)*				0.32 (3.0)*
SPJ	-0.56 (8.0)*				-0.80 (6.0)*
Training	1.79 (14.3)*				-0.05 (0.5)
Loglikelihood	-14,207.5	-1,800	-1,676.8	-1,166.0	-9,799.3

Table 2 Estimation results three districts; males, no unobserved heterogeneity (t-values in parentheses)

*: At a 5%-level the coefficient is significantly different from zero.

For the transition rate to a PUJ the only personal characteristic that has a coefficient significantly different from zero is the dummy for secondary and higher education. As was to be expected lower educated unemployed workers have a higher transition rate into PUJ. There is positive duration dependence in the transition rate to PUJ. The transition rate in the first six months is substantially below the one in later periods. SPJ and training have higher entrance rates for younger, higher educated and married unemployed. In many respects the transition rate to SPJ and training are complementary to the transition rate to a regular job.

The job separation rate turns out to be smaller for young workers. Furthermore, the job separation rate decreases with the educational level. And, married workers have a lower job separation rate. With the exception of a higher job separation rate between 6 and 12 months of job tenure there is no duration dependence in the job separation rate. The treatment effect of a PUJ turns out to be positive. This would imply that accepting a PUJ both stimulates the transition rate from unemployment to a job and the job separation rate. So, workers on a PUJ leave unemployment more quickly but also return to unemployment more quickly than their counterparts do. The opposite holds for SPJ. Once unemployed enter a SPJ their transition rate to a regular job decreases, but if they enter a regular job they stay on that job longer. Of course, these results are obtained under the assumption of no presence of unobserved heterogeneity, which also implies that we did not correct for selectivity in the inflow to the ALMP programs.

The estimation results for the second model are shown in Table 3. There appear to be unobserved determinants in three of the five transition rates. There is no unobserved heterogeneity in the transition to a SPJ or to training. With the introduction of unobserved heterogeneity the estimation results for the other transition rates improve, which is easy to see because within each of the transition rates the mass points are significantly different from each other at conventional levels of significance⁹. Within the transition rate from unemployment to a regular job there is conditional on the effects of personal characteristics and the elapsed duration of unemployment a small group of 5% that has a substantially lower transition rate than the majority has. The transition rate of this small group is only about 20% of the transition rate of the larger group. Within the transition rate to a PUJ there are also two groups. About two-thirds has a positive transition rate to a PUJ and about one-third has a transition rate of zero. Apparently, there is a group of workers that has no possibility to enter a PUJ. Finally, within the job separation

⁹ A formal Likelihood-Ratio test is problematic since one of the parameters (p_j) is not identified in the model without unobserved heterogeneity. The difference in the values of the loglikelihoods between the models with and without unobserved heterogeneity is large for the transition rate to a regular job (23.9) and for the job separation rate (38.8). The difference for the transition rate to PUJ is smaller (3.0), but here the probability does not go to an extreme value, while one of the mass points is well defined and the other goes to -∞. Therefore, here too unobserved heterogeneity seems to be present.

process there are two groups of about equal size. The differences in job separation rate between these two groups are substantial. One group has a transition rate from employment back to unemployment that is about 10% of the transition rate of the other group.

Table 3 Estimation results three districts; males, with unobserved heterogeneity in each of the transition rates (t-values in parentheses)

	To job	To PUJ	From job
Age <30	-0.05 (1.1)	-0.18 (1.2)	-0.01 (0.1)
Inc. Sec. ed.	0.21 (3.4)*	-0.33 (1.4)	-0.17 (2.0)*
Sec higher educ.	0.15 (3.3)*	-0.46 (2.5)*	-0.27 (4.0)*
Married	0.17 (4.1)*	-0.36 (2.5)*	-0.16 (2.4)*
District 2	-0.10 (2.5)*	-0.06 (0.4)	0.01 (0.1)
District 3	-0.53 (9.7)*	-0.25 (1.3)	0.27 (3.6)*
Mass points			
u ^a , v ^a , w ^a	-2.40 (39.8)*	-6.22 (20.4)*	-4.28 (26.4)*
u^b, v^b, w^b	-4.49 (5.3)*	-∞ (-)	-2.07 (20.9)*
Duration depend	lence		
3-4 Quarters	-0.22 (4.9)*	1.76 (6.4)*	1.45 (21.9)*
4-8 Quarters	-0.27 (4.8)*	2.74 (10.6)*	1.24 (8.7)*
8+ Quarters	-0.02 (0.2)	3.75 (13.5)*	1.54 (9.4)*
Treatment effect	s		
PUJ	0.43 (4.1)*		0.24 (1.4)
SPJ	-0.35 (3.1)*		-1.58 (8.7)*
Training	1.79 (12.3)*		-0.00 (0.0)
Unobs. het. prob	ability		
pj	0.95 (28.4)*	0.66 (9.4)*	0.48 (13.6)*
Loglikelihood	-14,183.6	-1,797.0	-9,760.5

The coefficients of the personal characteristics are hardly affected. The main changes in the parameter estimates refer to the coefficients of the duration dependence parameters and the coefficients that represent the treatment effects. In the estimates with unobserved heterogeneity the transition rate from unemployment to a job goes up after two years and is even higher than in the first six months in stead of being lower as in the previous estimates. The duration dependence of the transition rate to PUJ is hardly affected by the introduction of

unobserved heterogeneity, but the job separation rate is. Whereas in the baseline model there was hardly any duration dependence, now the job separation rate in the first six months is significantly smaller than it is in later periods.

Table 4 Estimation results three districts – correlated unobserved heterogeneity; males (t-values in parentheses)^{a)}

	To job	To PUJ	To SPJ	To training	From job
Age <30	-0.07 (1.5)	-0.07 (0.5)	0.24 (1.5)	0.44 (2.4)*	0.00 (0.0)
Inc Sec. ed.	0.21 (3.3)*	-0.36 (1.5)	0.74 (3.8)*	0.47 (2.0)*	-0.22 (2.7)*
Sec higher educ.	0.17 (3.6)*	-0.61 (3.4)*	0.61 (3.9)*	0.70 (3.8)*	-0.30 (4.5)*
Married	0.18 (4.1)*	-0.40 (2.6)*	0.45 (2.9)*	0.62 (3.3)*	-0.19 (2.9)*
District 2	-0.11 (2.5)*	-0.03 (0.2)	-0.10 (0.7)	-0.11 (0.6)	0.01 (0.1)
District 3	-0.56 (9.9)*	-0.23 (1.2)	0.28 (1.5)	0.16 (0.8)	0.29 (3.8)*
Mass points					
u ^a , v _a ^a , w ^a	-2.36 (40.0)*	-6.01 (19.6))* -∞(-)	-6.64 (18.8)*	-4.23 (29.4)*
$u^b, v_a{}^b, w^b$	-4.35 (10.2)*	-∞(-)	-4.73 (22.4)*	-6.90 (17.0)*	-1.93 (19.0)*
Duration depend	ence				
3-4 Quarters	-0.20 (4.4)*	1.72 (5.8)*	• -0.37 (2.3)*	0.42 (2.1)*	1.51 (21.8)*
4-8 Quarters	-0.22 (3.9)*	2.67 (9.6)*	• -0.51 (2.9)*	0.59 (3.0)*	1.27 (9.6)*
8+ Quarters	0.12 (1.1)	3.64 (12.5)	*-0.76 (2.7)*	0.59 (2.2)*	1.51 (10.4)*
Treatment effects	5				
PUJ	0.59 (3.8)*				-0.53 (2.7)*
SPJ	-0.45 (3.5)*				-0.26 (1.6)
Training	1.80 (10.5)*				-0.09 (0.4)
Unobs. het. proba	ability				
p ₁	0.07 (2.4)*		p ₅	0.38 (10.2)*	
p ₂	0.0 (-)		p_6	0.06 (3.1)*	
p ₃	0.45 (10.3)*		p ₇	0.02 (-)	
p ₄	0.02 (1.8)		p_8	0.0 (-)	
Loglikelihood		-28,55	59.5		

The introduction of unobserved heterogeneity hardly affects the treatment effects in the transition from unemployment to a job. However, there is a substantial change in the PUJ-treatment effect from a job back to unemployment. Whereas in the baseline model this treatment effect was significantly positive it is now positive, but not significantly different from zero. Of course, these results are obtained under the assumption that there is no correlation between the unobserved components.

Table 4 shows the estimation results for the third model in which the unobserved components are allowed to be correlated, so it allows for selectivity in the inflow into ALMP. In the estimation the unobserved components of the three ALMP transition rates are assumed to be perfectly correlated. Now, it appears that there is unobserved heterogeneity in the SPJ transition rate, which was not found in the previous estimates. Furthermore, it appears that in terms of unobserved components workers either have a positive PUJ transition rate or a positive SPJ transition rate. The estimation results indicated that there is selectivity in the inflow into ALMP¹⁰.

Two probabilities out of the eight that are distinguished go to zero so a discrete distribution with six points of support remains¹¹. The largest group has a share of 45% and consists of workers that have a higher transition rate from unemployment to a job, a zero transition rate to a PUJ, positive transition rates to SPJ and training and a low transition rate from a regular job back to unemployment than there counterparts have.

Conditional on the personal characteristics and the time spent in unemployment or employment this is the group of workers that has the best labor market characteristics. They find a job quickly and once they have found it they stick to it. The second largest group has a share of 38%. This is the group that has a high transition rate from unemployment to a regular job, a positive transition rate to a PUJ and training, a zero transition rate to a SPJ and a high job separation rate. So, this group leaves unemployment relatively quickly, but also returns to unemployment relatively quickly.

If we look at the unobserved heterogeneity distributions of each of the transition rates separately we see that they are different from those in Table 3. Within the job finding rate we now have a group of 92% that has a high value and 8% that has a low value. Within the transition rate to a PUJ we have a group of 47% with a

¹⁰ The sum of the loglikelihood-values in Table 3 and those of the SPJ transition rate and training transition rate in Table 2 is -28,583.9, which means that there is a difference in loglikelihood-value of 24.4 between the model with correlated unobserved heterogeneity and the one with independent unobserved heterogeneity. Again, no formal Likelihood-Ratio test can be applied. However the difference between the two likelihood-values seems large enough to reject the hypothesis of independent unobserved heterogeneity.

¹¹ This means that two out of the possible eight combinations we distinguish do not exist.

positive value and 53% with a zero value, meaning that this group has no probability to enter a PUJ. Finally, within the job separation rate we have groups of about equal size, 54% with a low value and 46% with a much higher value.

Comparing the coefficients in the models with independent and with correlated unobserved heterogeneity it is obvious that the estimates of the coefficients of the personal characteristics are hardly affected. Also, the pattern of duration dependence is approximately the same. What changes mostly are the treatment effects. For example, there is a negative correlation between the unobserved components of the transition rate to a regular job and the transition rate to a PUJ. This means that conditional on other determinants the groups that have a lower transition rate to a job have a higher transition rate to PUJ. If this negative correlation is not accounted for the effect a PUJ has on the transition rate to a regular job is underestimated. Therefore, the treatment effect as measured according to the third estimates is larger than the one we measured in the first two models. Overall there is a positive correlation between the unobserved components in the transition to a PUJ and the separation rate. This means that conditional on the other determinants workers that enter a PUJ are more likely to have short job tenures. If this correlation is not accounted for the effect of a PUJ on the job separation rate is overestimated. In fact, as shown in Table 4 the treatment effect is now negative: those workers that have been on a PUJ are more likely to have long job tenures than the workers that have not been on a PUJ. The positive effect of a PUJ on the exit rate out of unemployment and the negative effect on the job separation rate could have to do with signaling. By accepting a PUJ the unemployed worker signals a positive attitude towards work. Apparently, this leads both to a higher job finding rate and to a better quality of the match between worker and job. If only the first effect would be observed it would be difficult to give a clear interpretation. Unemployed that have been on a PUJ could lower their reservation wage for a regular job or increase their search intensity because they not want to stay on the PUJ. Now that a PUJ also effects the job tenure I think both treatment effects are related to employers' recruitment behavior.

With respect to the job finding rate the treatment effect of SPJ is negative irrespective of the model used. Apparently, workers that enter a SPJ reduce their efforts to find a regular job. This may have to do with the long-term character of a SPJ, which makes it from the perspective of the individual unemployed worker a good substitute for a regular job. The job separation rate is not affected by a previous SPJ-position, which indicated that workers that take a SPJ neither benefit from it nor are they stigmatized because of it.

These results with respect to SPJ and PUJ seem to be counterintuitive since PUJ were focused on disadvantaged groups while SPJ were targeted on unemployed with favorable labor market prospects. However, in the empirical model the explanatory variables account for some of the differences with respect to labor market prospects. And, perhaps the SPJ were too much like regular jobs. The PUJ

effects could have to do with signaling. The workers on PUJ may have unfavorable labor market characteristics, by accepting a PUJ the unemployed worker signals a positive attitude towards work.

The estimation results in Table 4 also indicate that training has a big impact on the job finding rate. Workers that enter a training program have a job finding rate that is 6 times as high as the job finding rate of other workers. I think this effect is unbelievably high and could indicate reversed causality: some unemployed only enter a training program after they have been promised a job¹². Finally, training does not make an unemployed more valuable to an employer since the job separation rate is not affected by previous participation in a training program.

5.2 Additional analysis

From the estimation results it is clear that the introduction of unobserved heterogeneity terms irrespective of whether they are correlated or not does not affect the estimates of the coefficients of the three transition rates very much. The main difference is in the measurement of the treatment effect of PUJ and SPJ on the job separation rate. In the baseline model there is no unobserved heterogeneity, a significant positive PUJ-treatment effect and a significant negative SPJ-treatment effect. In the second model with uncorrelated unobserved heterogeneity components an insignificant positive PUJ-treatment is found while the SPJ-treatment effect remains significantly negative. In the third model with correlation between the unobserved heterogeneity components I find a significant negative PUJ-treatment effect while the SPJ-treatment effect does not differ significantly from zero.

PUJ seem to be the most efficient ALMP. Workers that have been on a PUJ have a larger job finding rate and a smaller job separation rate. In the study by Bonnal et al. (1997) it is shown that the specification of the distribution of unobserved heterogeneity may affect the estimation results. To illustrate the importance of a flexible specification of unobserved heterogeneity I also estimated a model in which perfect correlation between the unobserved heterogeneity components is imposed. This means that in stead of six groups there are two groups of workers that conditional on the observed characteristics and the elapsed duration of unemployment or employment are homogeneous within but different between.

Imposing two-point heterogeneity I find a group of 56% that has a positive transition rate to a PUJ, a zero transition rate to SPJ and a small job separation rate. The other group of 44% has a zero transition rate to a PUJ, a positive transition rate to SPJ and a much higher job separation rate. Both groups hardly differ in their transition rate to training and their job finding rate. The effect on the

¹² Note that this also violates the non-anticipation condition necessary to avoid biased estimates of the impact of ALMPs on unemployment durations.

job finding rate of the ALMP programs is not much different than before. However the effects that a PUJ and a SPJ have on the job separation rate are different than before. The job separation PUJ-treatment effect is opposite to the effect as measured in the model without perfect correlation of the unobserved heterogeneity components whereas unlike before a SPJ now has a significant negative effect on the job separation rate. Apparently, if too heavy restrictions are imposed on the distribution of unobserved heterogeneity estimated treatment effects may be severely biased.

	Type a		Type b		Average	
Job finding	No PUJ	PUJ	No PUJ	PUJ	No PUJ	PUJ
After 6 months	50	50	90	90	54	54
After 1 year	28	14	83	75	34	21
After 2 year	16	4	76	62	23	10
Job separation						
After 6 months	93	96	49	65	72	81
After 1 year	66	78	2	9	35	45
After 2 years	51	66	0	2	27	35

Table 5 Simulation results, effects of a PUJ; survival probabilities (%)^a

a) Young, low educated, married worker in the district Bratislava; the PUJ starts after 6 months

To illustrate the size of the effects of a PUJ some simulations results are shown in Table 5. The simulations refer to a young, low educated, married worker in the district of Bratislava. There is a distinction between type a and type b workers. As shown in Table 5 type a workers have a survival probability after 6 months of 50% and a survival probability after 1 year of 28%. If a type a worker enters a PUJ at 6 months of unemployment his survival probability after 1 year is 14%. So, a PUJ has a big impact of the survival probabilities of unemployed workers. Type b workers have a much weaker labor market position. Their survival probability after 1 year is 83%. If a type b worker enters a PUJ at 6 months his survival probability after 1 year equals 75%. Here too, the effect is substantial. For the (weighted) average of the two types of workers a similar impact of a PUJ is found. The average survival probability after 6 months is 54% and after 1 year it is 34%. If a worker enters a PUJ at six months his average survival probability after 1 year is 21%. Table 5 also shows the treatment effect of a PUJ on the job separation rate. The two types of workers differ substantially in their survival rate. Of type a workers 93% still has his job after 6 months, of type b workers this is 49%. After

1 year of type a workers 66% still has his job, while of type b workers only 2% still has his job. On average after 1 year 35% of the workers still has his job, which is 45% for those workers that have been on a PUJ. Again, the treatment effect is substantial.

6 Conclusions

This paper studies job finding and job separation rates of male workers in three districts in the Slovak Republic. Job finding rates appear to be higher for higher educated and married workers. Job separation rates decrease with the educational level and are lower for married workers.

The main interest is in the effect that ALMP-programs have on job finding and job separation rates. PUJ, SPJ and training programs are used as a treatment to stimulate unemployed workers to find a regular job. It is important to evaluate whether the treatment actually works. The main issue in an evaluation study of this kind is the possible endogeneity of the inflow into the treatment program. It could be that unemployed workers that flow into an ALMP program have unobserved characteristics that are different from those that do not. It could be that workers that go to an ALMP are less able to find a regular job than otherwise identical non-participants in an ALMP are. Then a straightforward comparison of the two groups in terms of job finding rates leads to an underestimation of the effect of an ALMP. It is also possible that workers that go to an ALMP are less able to keep a regular job than otherwise identical non-participants in an ALMP are. Then a straightforward comparison of the two groups in terms of job separation rate leads to an overestimation of the effect of an ALMP. To account for this type of selectivity transition rate models are estimated that allow for the three transition rates to be interdependent.

The transition rate into a PUJ, a SPJ or training, the transition rate from unemployment to a regular job and the job separation rate may be influenced by unobserved variables that could be correlated. One of the findings of the paper is that it is very important to use a flexible specification of the correlation structure. SPJ appear to have a negative effect on the job finding rate and no effect on the job separation rate. The positive effect found for training on the job finding rate of unemployed workers may have to do with reversed causality: some workers enter a training program only after they are promised a job. Training does not seem to affect the job separation rate. PUJ seem to be the most efficient active labor market policy. Workers that are or have been on a PUJ have a higher job finding rate than other unemployed workers have and once they find a job they have a lower job separation rate than workers that have not been on a PUJ. These effects could have to do with signaling. By accepting a PUJ the unemployed worker signals a positive attitude towards work. Apparently, this leads both to a higher job finding rate and to a better quality of the match between worker and job.

References

Abbring, J.H., Van den Berg, G.J., Van Ours, J.C. (1997): "The Effect of Unemployment Insurance Sanctions on the Transition Rate from Unemployment to Employment". Working Paper Tinbergen Institute, Amsterdam/Rotterdam.

Abbring, J.H., Van den Berg, G.J. (1999): "The Non-Parametric Identification of Treatment Effects in Duration Models". mimeo, Vrije Universiteit Amsterdam.

Boeri, T. (1997): "Labor-market Reforms in Transition Economies". Oxford Review of Economic Policy, 13, 126-140.

Bonnal, L., Fougère, D., Sérandon A. (1997): "Evaluating the Impact of French Employment Policies on Individual Labor Market Histories". Review of Economic Studies, 64, 683-713.

Gritz, R.M. (1993): "The Impact of Training on the Frequency and Duration of Employment". Journal of Econometrics, 57, 21-51.

Heckman, J.J., LaLonde, R.J., Smith, J.A. (1999): "The Economics and Econometrics of Active Labor Market Programs". in: Ashenfelter O., Card, D. (eds.), Handbook of Labor Economics, volume III, North-Holland.

Holm, A., Van den Berg, G.J., Van Ours, J.C. (1999): "Does work experience help to become a medical specialist". Discussion Paper 9919, CentER for Economic Research, Tilburg University.

Lubyova, M., Van Ours, J.C. (1997): "Unemployment Dynamics and the Restructuring of the Slovak Unemployment Benefit Systems". European Economic Review, 41, 923-934.

Lubyova, M., Van Ours, J.C. (1998): "Work Incentives and Other Effects of the Transition to Social Assistance: Evidence from the Slovak Republic". Empirical Economics, 23, 1/2: 121-154.

Lubyova, M., Van Ours, J.C. (1999): "Effect of Active Labor Market Programs on the Transition Rate from Unemployment into Regular Jobs in the Slovak Republic". Journal of Comparative Economics, 27, 90-112.

Organization for Economic Co-operation and Development (1996): Labor Markets and Social Policies in the Slovak Republic, Paris: OECD

Puhani, P.A. (1998): "Advantage through Training? A Microeconomic Evaluation of the Employment Effects of Active Labor Market Programs in Poland". ZEW-Discussion Paper no. 98-25, Mannheim.

Ridder, G. (1986): "An event history approach to the evaluation of training, recruitment, and employment programs". Journal of Applied Econometrics, 11, 109-126.

Svejnar, J. (1999): Labor Markets in the Transitional Central and East European Economies". in: Ashenfelter O., Card, D. (eds.), Handbook of Labor Economics, volume III, North-Holland.

Van den Berg, G.J., Van der Klaauw, B., Van Ours, J.C. (1998): "Punitive Sanctions and the Transition Rate from Welfare to Work", *Discussion Paper* CentER for Economic Research no. 9856, Tilburg University.

Van den Berg, G.J. (2000): "Duration Models: Specification, Identification, and Multiple Durations", in: Heckman, James J., and Leamer, Edward (eds.), *Handbook of Econometrics*, Volume V, North-Holland, forthcoming.

Appendix 1- Definition and means of variables of the 1993 inflow sample from the three Slovak districts; males

All variables are dummy variables (with value 1 where indicated and value 0 for the reference group)

Definitions of variables

Age<30: age is below 30 years as measured in 1998

Reference group age: age is 30 years or more in 1998

Incomplete secondary education

Secondary and higher education (including university)

Reference group education: no education - basic education - apprentice

Married: married or cohabiting person

Reference group marital status: single-divorced - widow/widower - unknown

Means of variables	Bratislava	a Dolny Kubin	Bardejo	ov Total
(Number of observations)	(1499)	(1902)	(1656)	(5057)
Age < 30 years	0.23	0.39	0.42	0.35
Age \geq 30 years	0.77	0.61	0.58	0.65
Lower education	0.72	0.72	0.32	0.59
Incomplete sec education	0.06	0.06	0.40	0.17
Sec and higher education	0.22	0.22	0.28	0.24
Married	0.70	0.58	0.46	0.58
Other marital status	0.30	0.42	0.54	0.42
Unemployment rate (%)	4.1	12.9	19.3	12.4

IZA Discussion Papers

No.	Author(s)	Title	Area	Date
21	J. Ondrich C. K. Spiess Q. Yang G. G. Wagner	The Liberalization of Maternity Leave Policy and the Return to Work after Childbirth in Germany	1/3	9/98
22	A. Million R. Rotte K. F. Zimmermann	The Economics of Hospital Trips and Health Care Reform	3	9/98
23	T. Bauer K. F. Zimmermann	Learning Efficiency of Economics Students	7	9/98
24	K. F. Zimmermann M. C. Burda J. von Hagen K. Konrad F. Schneider G. G. Wagner	Petersberger Erklärung: Anstöße für eine zukunftsgerichtete Arbeitsmarktpolitik	1/3	10/98
25	K. Papps R. Winkelmann	Unemployment and Crime: New Answers to an Old Question	3/7	11/98
26	D. Enste F. Schneider	Increasing Shadow Economies All Over the World - Fiction or Reality?	3/7	12/98
27	E. Bratberg Ø. A. Nilsen	Transitions from School to Work: Search Time and Job Duration	7	12/98
28	U. Broll K. P. Wong	Efficiency Wages and Futures Markets	1/7	1/99
29	D. J. Snower	Causes of Changing Earnings Inequality	2/3	1/99
30	J. Kluve H. Lehmann C. M. Schmidt	Active Labor Market Policies in Poland: Human Capital Enhancement, Stigmatization or Benefit Churning?	4	1/99
31	C. M. Schmidt	Persistence and the German Unemployment Problem: Empirical Evidence on German Labor Market Flows	1/7	2/99
32	S Å. Dahl Ø. A. Nilsen K. Vaage	Work or Retirement? Exit Routes for Norwegian Elderly	3/7	2/99
33	A. Lindbeck D. J. Snower	Price Dynamics and Production Lags	1/7	2/99
34	P. A. Puhani	Labour Mobility – An Adjustment Mechanism in Euroland?	1/2	3/99

35	D. A. Jaeger A. Huff Stevens	Is Job Stability in the United States Falling? Reconciling Trends in the Current Population Survey and Panel Study of Income Dynamics	1	3/99
36	C. Lauer	The Effects of European Economic and Monetary Union on Wage Behaviour	1/2	3/99
37	H. S. Buscher C. Müller	Exchange Rate Volatility Effects on the German Labour Market: A Survey of Recent Results and Extensions	1/2	3/99
38	M. E. Ward P. J. Sloane	Job Satisfaction within the Scottish Academic Profession	7	4/99
39	A. Lindbeck D. J. Snower	Multi-Task Learning and the Reorganization of Work	1/5	4/99
40	S. M. Golder T. Straubhaar	Empirical Findings on the Swiss Migration Experience	1	4/99
41	J. M. Orszag D. J. Snower	Anatomy of Policy Complementarities	3/7	5/99
42	D. S. Hamermesh	The Changing Distribution of Job Satisfaction	7	5/99
43	C. Belzil J. Hansen	Household Characteristics, Ability and Education: Evidence from a Dynamic Expected Utility Model	7	5/99
44	D. N. F. Bell R. A. Hart	Overtime Working in an Unregulated Labour Market	1	6/99
45	R. A. Hart J. R. Malley	On the Cyclicality and Stability of Real Earnings	1	6/99
46	R. Rotte M. Vogler	The Effects of Development on Migration: Theoretical Issues and New Empirical Evidence	2	6/99
47	R. A. Hart	Tenure-based Wage Setting	1/7	7/99
48	T. Bauer K. F. Zimmermann	Overtime Work and Overtime Compensation in Germany	1	7/99
49	H. P. Grüner	Unemployment and Labor-Market Reform: A Contract Theoretic Approach	1/3	7/99
50	K. F. Zimmermann	Ethnic German Migration After 1989 – Balance and Perspectives	1	8/99
51	A. Barrett P. J. O'Connell	Does Training Generally Work? The Returns to In-Company Training	7	8/99
52	J. Mayer R. T. Riphahn	Fertility Assimilation of Immigrants: Evidence from Count Data Models	3	8/99
53	J. Hartog P. T. Pereira J. A. C. Vieira	Inter-industry Wage Dispersion in Portugal: high but falling	7	8/99
54	M. Lofstrom	Labor Market Assimilation and the Self-Employment Decision of Immigrant Entrepreneurs	1	8/99

55	L. Goerke	Value-added Tax versus Social Security Contributions	3	8/99
56	A. Lindbeck D. J. Snower	Centralized Bargaining and Reorganized Work: Are they compatible?	1/5	9/99
57	I. N. Gang K. F. Zimmermann	Is Child like Parent? Educational Attainment and Ethnic Origin	1	9/99
58	T. Bauer K. F. Zimmermann	Occupational Mobility of Ethnic Migrants	1	9/99
59	D. J. DeVoretz S. A. Laryea	Canadian Immigration Experience: Any Lessons for Europe?	1/2/3	9/99
60	C. Belzil J. Hansen	Subjective Discount Rates, Intergenerational Transfers and the Return to Schooling	7	10/99
61	R. Winkelmann	Immigration: The New Zealand Experience	7	10/99
62	A. Thalmaier	Bestimmungsgründe von Fehlzeiten: Welche Rolle spielt die Arbeitslosigkeit?	3	10/99
63	M. Ward	Your Everyday, Average Academic	5	10/99
64	M. Ward	Salary and the Gender Salary Gap in the Academic Profession	5	10/99
65	H. Lehmann J. Wadsworth A. Acquisti	Grime and Punishment: Job Insecurity and Wage Arrears in the Russian Federation	4	10/99
66	E. J. Bird H. Kayser J. R. Frick G. G. Wagner	The Immigrant Welfare Effect: Take-Up or Eligibility?	3	10/99
67	R. T. Riphahn A. Thalmaier	Behavioral Effects of Probation Periods: An Analysis of Worker Absenteeism	1/3	10/99
68	B. Dietz	Ethnic German Immigration from Eastern Europe and the former Soviet Union to Germany: the Effects of Migrant Networks	1	11/99
69	MS. Yun	Generalized Selection Bias and the Decomposition of Wage Differentials	7	11/99
70	I. N. Gang F.L. Rivera-Batiz	Immigrants and Unemployment in the European Community	1	11/99
71	L. Goerke	The Wedge	3	11/99
72	J. Fersterer R. Winter-Ebmer	Are Austrian Returns to Education Falling Over Time?	7	11/99
73	G. S. Epstein S. Nitzan	The Endogenous Determination of Minimum Wage	3	11/99

74	M. Kräkel	Strategic Mismatches in Competing Teams	7	12/99
75	B. Henry M. Karanassou D. J. Snower	Adjustment Dynamics and the Natural Rate: An Account of UK Unemployment	1	12/99
76	G. Brunello M. Giannini	Selective Schools	7	12/99
77	C. M. Schmidt	Knowing What Works: The Case for Rigorous Program Evaluation	6	12/99
78	J. Hansen R. Wahlberg	Endogenous Schooling and the Distribution of the Gender Wage Gap	7	12/99
79	J. S. Earle Z. Sakova	Entrepreneurship from Scratch: Lessons on the Entry Decision into Self-Employment from Transition Economies	4	12/99
80	J. C. van Ours J. Veenman	The Netherlands: Old Emigrants – Young Immigrant Country	1	12/99
81	T. J. Hatton S. Wheatley Price	Migration, Migrants and Policy in the United Kingdom	1	12/99
82	K. A. Konrad	Privacy, time consistent optimal labor income taxation and education policy	3	12/99
83	R. Euwals	Female Labour Supply, Flexibility of Working Hours, and Job Mobility in the Netherlands	1	12/99
84	C. M. Schmidt	The Heterogeneity and Cyclical Sensitivity of Unemployment: An Exploration of German Labor Market Flows	1	12/99
85	S. Pudney M. A. Shields	Gender and Racial Discrimination in Pay and Promotion for NHS Nurses	5/6	12/99
86	J.P. Haisken-DeNew C. M. Schmidt	Money for Nothing and Your Chips for Free? The Anatomy of the PC Wage Differential	5/7	12/99
87	T. K. Bauer	Educational Mismatch and Wages in Germany	7	12/99
88	O. Bover P. Velilla	Migration in Spain: Historical Background and Current Trends	1	12/99
89	S. Neuman	Aliyah to Israel: Immigration under Conditions of Adversity	1	12/99
90	H. Lehmann J. Wadsworth	Tenures that Shook the World: Worker Turnover in Russia, Poland and Britain	4	12/99
91	M. Lechner	Identification and Estimation of Causal Effects of Multiple Treatments Under the Conditional Independence Assumption	6	12/99
92	R. E. Wright	The Rate of Return to Private Schooling	7	12/99
93	M. Lechner	An Evaluation of Public-Sector-Sponsored Continuous Vocational Training Programs in East Germany	6	12/99

94	M. Eichler M. Lechner	An Evaluation of Public Employment Programmes in the East German State of Sachsen-Anhalt	6	12/99
95	P. Cahuc A. Zylberberg	Job Protection, Minimum Wage and Unemployment	3	12/99
96	P. Cahuc A. Zylberberg	Redundancy Payments, Incomplete Labor Contracts, Unemployment and Welfare	3	12/99
97	A. Barrett	Irish Migration: Characteristics, Causes and Consequences	1	12/99
98	J.P. Haisken-DeNew C. M. Schmidt	Industry Wage Differentials Revisited: A Longitudinal Comparison of Germany and USA	5/7	12/99
99	R. T. Riphahn	Residential Location and Youth Unemployment: The Economic Geography of School-to-Work- Transitions	1	12/99
100	J. Hansen M. Lofstrom	Immigrant Assimilation and Welfare Participation: Do Immigrants Assimilate Into or Out-of Welfare?	1/3/7	12/99
101	L. Husted H. S. Nielsen M. Rosholm N. Smith	Employment and Wage Assimilation of Male First Generation Immigrants in Denmark	3	1/00
102	B. van der Klaauw J. C. van Ours	Labor Supply and Matching Rates for Welfare Recipients: An Analysis Using Neighborhood Characteristics	2/3	1/00
103	K. Brännäs	Estimation in a Duration Model for Evaluating Educational Programs	6	1/00
104	S. Kohns	Different Skill Levels and Firing Costs in a Matching Model with Uncertainty – An Extension of Mortensen and Pissarides (1994)	1	1/00
105	G. Brunello C. Graziano B. Parigi	Ownership or Performance: What Determines Board of Directors' Turnover in Italy?	7	1/00
106	L. Bellmann S. Bender U. Hornsteiner	Job Tenure of Two Cohorts of Young German Men 1979 - 1990: An analysis of the (West-)German Employment Statistic Register Sample concerning multivariate failure times and unobserved heterogeneity	1	1/00
107	J. C. van Ours G. Ridder	Fast Track or Failure: A Study of the Completion Rates of Graduate Students in Economics	7	1/00
108	J. Boone J. C. van Ours	Modeling Financial Incentives to Get Unemployed Back to Work	3/6	1/00
109	G. J. van den Berg B. van der Klaauw	Combining Micro and Macro Unemployment Duration Data	7	1/00
110	D. DeVoretz C. Werner	A Theory of Social Forces and Immigrant Second Language Acquisition	1	2/00

111	V. Sorm K. Terrell	Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic	1/4	2/00
112	L. Bellmann T. Schank	Innovations, Wages and Demand for Heterogeneous Labour: New Evidence from a Matched Employer-Employee Data-Set	5	2/00
113	R. Euwals	Do Mandatory Pensions Decrease Household Savings? Evidence for the Netherlands	7	2/00
114	G. Brunello A. Medio	An Explanation of International Differences in Education and Workplace Training	7	2/00
115	A. Cigno F. C. Rosati	Why do Indian Children Work, and is it Bad for Them?	7	2/00
116	C. Belzil	Unemployment Insurance and Subsequent Job Duration: Job Matching vs. Unobserved Heterogeneity	7	2/00
117	S. Bender A. Haas C. Klose	IAB Employment Subsample 1975-1995. Opportunities for Analysis Provided by the Anonymised Subsample	7	2/00
118	M. A. Shields M. E. Ward	Improving Nurse Retention in the British National Health Service: The Impact of Job Satisfaction on Intentions to Quit	5	2/00
119	A. Lindbeck D. J. Snower	The Division of Labor and the Market for Organizations	5	2/00
120	P. T. Pereira P. S. Martins	Does Education Reduce Wage Inequality? Quantile Regressions Evidence from Fifteen European Countries	5/7	2/00
121	J. C. van Ours	Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?	4/6	3/00